RESEARCH ARTICLE

Open Access

Prevalence, features and health impacts of eating disorders amongst First-Australian Yiramarang (adolescents) and in comparison with other Australian adolescents



Adam Burt¹, Deborah Mitchison^{2,3*}, Elizabeth Dale⁴, Kay Bussey⁵, Nora Trompeter⁵, Alexandra Lonergan⁵ and Phillipa Hay^{2,6}

Abstract

Background: This study aimed to support previous research conducted with First-Australians (FA) by establishing the prevalence of eating disorders, and their demographic distribution and burden in adolescent First-Australians compared to other-Australians (OA).

Methods: Data were used from the baseline survey of the EveryBODY Study, a longitudinal investigation of eating disorders among Australian adolescents. Of the 5068 participants included, 402 (8%) identified as FA, 4586 (90.5%) identified as OA. Diagnosis of eating disorders was based on the Diagnostic and Statistical Manual version 5. Socioeconomic status and measures of impairment were assessed using validated instruments. Body mass index was calculated based on self-reported weight and height. Statistical analyses used data weighted to the distribution of gender in adolescents in New South Wales in the 2016 Australian Census. Chi-square tests were performed to determine prevalence of eating disorders amongst FA and to compare to OA. ANOVA and logistic regression analyses where conducted to examine the moderation effect of sociodemographic status, measures of impairment and FA status on the distribution of eating disorders.

Results: The prevalence rates for eating disorder diagnoses where similar for FA and OA with the exception of Night eating Syndrome (OSFED-NES), which occurred in 7.14% (95%CI 4.81–10.49) of FA vs. 3.72% (95%CI 3.17–4.36) in OA. The greater prevalence of OSFED-NES in FA was largely explained by poorer psychosocial quality of life amongst FA. (Continued on next page)

University, North Ryde, New South Wales, Australia

Full list of author information is available at the end of the article



^{*} Correspondence: deborah.mitchison@westernsydney.edu.au

²Translational Health Research Institute (THRI), School of Medicine, Western Sydney University, Sydney, Campbelltown, Australia ³Centre for Emotional Health, Department of Psychology, Macquarie

(Continued from previous page)

Conclusion: Eating disorders are common amongst First-Australian adolescents and are associated with poor psychosocial quality of life. These findings are consistent with previous research conducted with First-Australian adults. There is a need to screen for eating disorders amongst First-Australian adolescent girls and boys.

Keywords: Feeding and eating disorders, Adolescents, Prevalence, Oceanic ancestry, Group, Aboriginal and Torres Strait islander

Plain English summary

This study aimed to support previous research conducted with First Australian adults by establishing the prevalence, distribution and burden of eating disorders amongst First Australian adolescents, and compare this to other Australian adolescents. Data were used from the EveryBODY study, a survey of eating disorders amongst Australian adolescents. Of 5068 adolescents who completed the survey, 8% (402) were First Australians. All participants answered questions on basic demographic information, eating disorders, and impairment. The prevalence of eating disorders amongst First Australian adolescents was 28.62%, which was significantly greater than among other-Australian adolescents (21.68%). The greater prevalence of eating disorders amongst First Australians was largely driven by the higher prevalence of Night Eating Syndrome (7.14% amongst Fist Australians vs. 3.72% amongst other Austalians) and poorer psychosocial quality of life.

Background

First-Australians have poorer physical and mental health than other-Australians in all age groups, including adolescents [1, 2]. First-Australian adolescents have a two-fold greater all-cause mortality rate than other-Australian adolescents, mostly due to suicide and road traffic injuries. Eighty percent of this mortality has been reported as potentially avoidable within the current Australian healthcare system [2]. First-Australian adolescents also have a greater burden of non-fatal disease [1]. Approximately 20% of First-Australian adolescents are obese and 40% are either obese or overweight, amongst the highest rates for adolescents in the world [2]. First-Australian adolescents aged 18-24 years suffer twice the rate of high to very high psychological distress of other-Australian adolescents [2], not surprisingly First-Australian adolescents also have a significantly greater burden of mental disorders including: conduct disorder, intentional self-harm, anxiety disorders, depressive disorders and alcohol use disorders [1]. Hospital separations for psychosis, alcohol and other substance use disorders are three-fold higher in First-Australian adolescents compared to other-Australian adolescents [2]. These statistics are alarming, as First-Australians are on average younger than other-Australians (21.8 years vs. 37.6 years). FirstAustralian adolescents also make up a greater proportion of their respective population compared to other-Australian adolescents (37.7% vs. 20.4%) [1, 2].

Adolescence is an important developmental period characterised by a range of physical, cognitive and psychosocial changes, [3] it is also the time when eating disorders are most likely to emerge [4]. Evidence from an Australian prospective cohort study suggests that the prevalence of eating disorders amongst Australian adolescents is 6.2% for one of the three major eating disorders (i.e., anorexia nervosa, bulimia nervosa, binge eating disorder) and as high as 22.2% when including atypical and subthreshold eating disorders [5]. Yet the prevalence of eating disorders amongst First-Australian adolescents remains unknown.

There is a lack of good quality research into mental disorders amongst First-Australians generally, [6, 7] but especially so in regards to eating disorders. A systematic review into the prevalence of mental disorders amongst First-Australians did not identify any studies that assessed eating disorders [7]. A systematic synthesis of First-Australian Adolescent health population data resorted to the use of hospital separation data in order to measure the prevalence of eating disorders, as well as other mental disorders, as there were insufficient data available from surveys. The latter study did not describe the prevalence of individual eating disorders [2]. The research that is available on eating disorders amongst First-Australians is conflicting. An example of this was seen in the previously mentioned systematic synthesis of First-Australian adolescent health population data, which suggested that eating disorders are more common amongst other-Australian compared to First-Australian adolescents and are more common amongst urban compared to rural First-Australian adolescents [2]. These data conflict with research suggesting eating disorders occur at similar rates across urbanicity and socioeconomic status [8, 9]. Further to this, emerging research indicates that eating disorders are actually more common amongst First-Australian compared to other-Australian adults and are associated with high levels of overvaluation of weight and shape concerns. The same study found the higher prevalence of eating disorders amongst First-Australians' was largely associated with the younger age, higher BMI and poorer mental health of First-Australians [10].

In support of this, previous Australian population health and body image research has also suggested that these explanatory variables (younger age, higher BMI and mental health impairment), as well as physical health impairment and body image disturbance are also greater amongst First-Australian compared to other-Australian adolescents. This line of research thus indicates the potential of elevated risk of eating disorders among First-Australian adolescents. For example, an Australian study into media influences on body image and disordered eating among First-Australian adolescents found that First-Australian adolescents were more dissatisfied with their weight and engaged in more activity to change their weight and shape compared to other-Australian adolescents. First-Australian adolescents in this study were also more sensitive to media messaging to lose weight, despite receiving fewer media messages [11]. It should be noted however that this study was conducted in the early 2000s and newer research is needed to determine the impacts of today's widespread personalised and multimedia messaging on First-Australian adolescents' perceptions.

Based on the recent research findings into eating disorders amongst First-Australian adults, we suspect that the relatively larger pool of young people in the First-Australian community, burdened with other risk factors such as body image disturbance, poorer mental and physical health and higher rates of overweight and obesity may place them at greater risk of eating disorders.

Gap and aim

It is important to investigate the prevalence of eating disorders amongst First-Australian adolescents, especially given the sparse evidence available and the evidence to date suggesting a potentially elevated risk of eating disorders among this group. Further, this line of research forms a priority area for the Australian Government. The National Aboriginal and Torres Strait Islander Health plan 2013-2023 recognises evidence-based practice as important to guide the health care of First-Australians and asserts a need for more high quality research conducted in partnership with First-Australian communities to achieve this. The plan highlights adolescent, youth and mental health as specific priority areas for research and improvements [12, 13]. Thus this study aimed to support previous research conducted with First-Australians by establishing the prevalence of eating disorders, and their demographic distribution and burden in adolescent First-Australians compared to other-Australians.

Methods

Procedures and participants

Data were used from the baseline survey of the Every-BODY Study. The EveryBODY study is a longitudinal investigation of eating disorders among Australian

adolescents. Sampling procedures have been detailed elsewhere [5, 14]. In brief, four independent and nine government schools, from a broad range of socioeconomic advantage, participated. All parents and students received information about the study over a period of four weeks using multiple methods of dissemination, and a passive parental consent procedure was used, whereby parents could opt out their child from the study. Students who provided consent were given the online survey to complete at school. Participants were offered the chance to enter a prize draw to win one of ten gift vouchers, and the schools received a general wellbeing report based on their students' data.

Of the 5280 students who commenced the survey, n = 123 were excluded due to: completion of < 10% of the survey (n = 39), non-serious responses to open-ended questions (n = 79), incorrect coding (n = 4) and withdrawn consent (n = 1). Of the remaining participants, n = 402 (8%) identified as First Australian (n = 355 of Aboriginal origin, n = 28 of Torres Strait Islander origin and 19 of both Aboriginal and Torres Strait Islander origin), n = 4586 (90.5%) identified as other-Australian, and n = 80 (1.6%) declined to provide a response to the question regarding Indigenous status.

Diagnostic status and measures

Sociodemographic questions

A socio-economic index for area (SEIFA) score was calculated based on the residential postcode of participants. A SEIFA score is an Australian Bureau of Statistics (ABS) product used to rank areas in Australia based on relative socioeconomic advantage or disadvantage [15]. Other demographic information collected included age, school grade, gender, sex, and country of birth.

Eating disorder diagnoses

The operationalisation of the eating disorder diagnoses that were assessed in this study has been published previously [5], and an adapted Table of this operationalisation is included in the Supplementary material. Diagnoses assessed included the three major eating disorders (anorexia nervosa, bulimia nervosa, binge eating disorder), the five Other Specified Feeding and Eating Disorders (OSFED; atypical anorexia nervosa, subthreshold bulimia nervosa, subthreshold binge eating disorder, purging disorder, night eating syndrome) and Unspecified Feeding and Eating Disorder (UFED). Most symptoms were captured by items of the Eating Disorder Examination Questionnaire (EDE-Q), which assesses the presence and severity of cognitive and behavioural eating disorder symptoms and features [16]. This questionnaire has previously been validated in Australian adolescent boys and girls and demonstrates sound reliability [17]. Items used in this study included the behavioural

frequency items (self-induced vomiting, laxative misuse, driven exercise, and binge eating), and the Likert-type items that comprise the combined weight and shape concern subscales. As the frequency of behaviours were only assessed over the past 1 month (not the 3 months duration required for bulimia nervosa and binge eating disorder), we use the term "probable" for these diagnoses. Cronbach's alpha for the combined weight and shape concern subscale in the present study was 0.96 for both First-Australian and other-Australian adolescents.

Participant's self-reported current weight and height, which was converted to age and gender adjusted body mass index (BMI) percentiles for children and adolescents. A BMI percentile < 10 was used for the underweight criterion of anorexia nervosa, as this cut-off has most frequently been used in adolescent epidemiological studies of DSM-5 anorexia nervosa [18–21]. Three items from the Night Eating Questionnaire (NEQ) [22] were used to assess symptoms of night eating syndrome, including proportion of daily food intake consumed following supper, nocturnal eating (eating after going to bed), and awareness during nocturnal eating. The NEQ has been validated in adolescents and is superior to parent report [23].

Several additional questions were developed by the researchers to capture frequency of additional extreme weight control behaviours (fasting, strict dieting, detoxes, insulin misuse, other drug use for weight loss), distress associated with binge eating, and additional diagnostic binge eating disorder features (e.g., eating faster than usual, eating alone due to embarrassment). Participants were also asked about any recent weight loss in the past 4 weeks to assess atypical anorexia nervosa.

Quality of life impairment

Scores from the Paediatric Quality of Life Scale (PedsQL) SF15 [24, 25] were used to measure quality of life impairment. The 12 items from the physical functioning, emotional functioning, and social functioning subscales were included in the survey. Items ask participants to indicate on a Likert type scale how true a series of statements are of them in the past 4 weeks. Scores are reversed and transformed on a 0-100 scale, such that higher scores indicate higher functioning. Subscale scores are derived as the mean of the items for that scale. For the purposes of this study we combined the emotional and social functioning scales to create a psychosocial subscale. The PedsQL SF15 has evidence of good reliability and validity in previous studies of adolescents [25]. Cronbach's alphas in the current study sample for the physical functioning subscale was 0.89 and 0.84 for First Australian and other-Australian adolescents respectively, and for the psychosocial functioning subscale was 0.91 and 0.90 for First-Australian and other-Australian adolescents respectively.

Statistical analysis

Analyses used data weighted to the distribution of gender in adolescents in New South Wales in the 2016 Australian Census. A series of Chi-square tests were performed to determine the prevalence of eating disorders among First-Australian adolescents and to compare this to other-Australian adolescents. This was followed by 2 × 2 ANOVAs and logistic regression analyses to examine the moderation effect of First-Australian status on the sociodemographic (dependent variables: age, sex, SEIFA score, BMI percentile) distribution of eating disorders. Additional 2 x 2 ANOVAs and logistic regressions were employed to test the moderation effect of First-Australian status on the association between eating disorders and specific symptoms (dependent variables: weight and shape concerns, binge eating, self-induced vomiting, laxative use, dietary fasting, driven exercise) and impairment (psychosocial quality of life and physical quality of life impairment). Finally, four logistic regressions were performed with current eating disorder status as the outcome. In regression one, First-Australian status was the only predictor in the model; in regression two, sociodemographic variables (age, gender, SEIFA score, and BMI percentile) were added to the model; in regression three, eating disorder symptom variables (weight and shape concerns, binge eating, self-induced vomiting, laxative use, fasting, driven exercise) were added to the model; in regression four, impairment variables (psychosocial and physical quality of life impairment) were added to the model. The intention of these analyses was to examine the extent to which First-Australian status remained independently associated with the likelihood to meet criteria for a current eating disorder in the presence of other known correlates. In these analyses, SEIFA score was divided by 100 (the normed SD), BMI percentile was divided by 10, and weight/shape concerns and psychosocial and physical quality of life impairment scores were standardised, to allow more meaningful interpretation of the adjusted odds ratios.

Results

Prevalence of eating disorders amongst first-Australian adolescents

Overall, the prevalence of any eating disorder (i.e., major eating disorder, OSFED or UFED) was higher among First-Australian adolescents (28.6%) compared to other Australian adolescents (21.7%). Of note, eating disorder prevalence was also very high among adolescents who chose not to specify their First-Australian status (32.8%), however possibly due to reduced power in this smaller sub-group, this proportion did not differ statistically

Burt et al. Journal of Eating Disorders (2020) 8:10 Page 5 of 10

from either the First Australian or other-Australian groups. Further analyses excluded participants who did not disclose their First-Australian status. Table 1 displays the point prevalence for each eating disorder, separated by First-Australian status. As can be seen, there was a higher prevalence of any eating disorder and OSFED in First-Australian adolescents compared to other adolescents. This seemed to be largely explained by a higher prevalence of night eating syndrome (7.1% vs 3.7%, respectively) and purging disorder (5.1% vs 3.1%, respectively), although the latter neared but did not reach statistical significance. Other specific eating disorder diagnoses also trended to be more prevalent among First-Australian adolescents however these differences did not reach or come close to statistical significance.

Demographic distribution of eating disorders as moderated by first-Australian status

As can be seen in Table 2, although the sex distribution was similar in First-Australian and other-Australian adolescents in this sample, on average First-Australian adolescents were slightly younger, from lower socioeconomic backgrounds, and had a higher BMI percentile than other-Australian adolescents. Adolescents with eating disorders in general were more likely to be female, slightly older, and have a higher BMI percentile. Analyses of the interaction between First-Australian status and eating disorder status revealed that First-Australian status was not a significant moderator of the relationship between current eating

disorder and age, sex, socioeconomic status, or BMI. These findings indicate that First-Australian and other-Australian adolescents with an eating disorder have similar sociodemographic distributions.

Clinical correlates of eating disorders and moderation by first-Australian status

As can be seen in Table 2, First-Australian adolescents in general reported greater physical quality of life impairment and more frequent self-induced vomiting but similar levels of other eating disorder symptoms and psychosocial quality of life. As expected, adolescents with eating disorders in general reported more eating disorder symptoms and greater impairment in quality of life than adolescents without an eating disorder. No moderating effects of First-Australian status were found, indicating that First-Australian and other-Australian adolescents with eating disorders report similar symptom profiles and impairment.

Multivariate model of eating disorder correlates

As can be seen in Table 3, First-Australian status was a significant univariate correlate of current eating disorder status, with First-Australian adolescents around 1.5 times more likely to meet criteria for a current eating disorder compared to other-Australian adolescents. This effect remained significant even when the model accounted for sociodemographic variables and for behavioural and cognitive eating disorder symptoms. However, when measures of impairment were added to the model, First-Australian status was no longer independently associated

Table 1 One-month prevalence of eating disorders in First-Australian (FA) and other-Australian (OA) adolescents

1 3			
	First-Australian Adolescents	Other-Australian Adolescents	
	% (95% CI)	% (95% CI)	χ^2 (1), p
Any Eating Disorder (Major, OSFED, UFED)	28.62 (23.83–33.94)	21.68 (20.38–23.03)	7.84, .005
Major Eating Disorder	7.53 (5.15–10.88)	6.07 (5.38–6.85)	.13, .288
Anorexia Nervosa (AN)	.89 (.30–2.59)	.66 (.45–.96)	0.49, .401†
Probable Bulimia Nervosa (BN)	5.11 (3.21–8.02)	4.45 (3.86–5.13)	.31, .579
Probable Binge Eating Disorder (BED)	1.80 (.83–3.87)	.96 (.70–1.31)	2.16, .142
Other Specified Feeding and Eating Disorders (OSFED)	15.22 (11.71–19.55)	10.77 (9.83–11.78)	5.97, .015
Atypical Anorexia Nervosa (OSFED-AAN)	3.92 (2.30–6.58)	2.69 (2.23–3.23)	1.71, .191
Subthreshold Bulimia Nervosa (OSFED-SBN)	1.50 (.64–3.47)	2.21 (1.8–2.71)	.74, .390
Subthreshold Binge Eating Disorder (OSFED-SBED)	.30 (.05–1.68)	.32 (.19–.55)	<.01, .952
Purging Disorder (OSFED-PD)	5.11 (3.21–8.02)	3.12 (2.63–3.7)	3.82, .051
Night Eating Syndrome (OSFED-NES)	7.14 (4.81–10.49)	3.72 (3.17–4.36)	9.12, .003
Unspecified Feeding or Eating Disorder (UFED)	5.32 (3.30–8.46)	4.47 (3.85–5.19)	.46, .500

Note. Significant effects are highlighted in bold text. Based on weighted data. † based on Fishers Exact test due to low cell numbers. Total N for prevalence analyses varied for each diagnosis, dependent on the missingness of diagnostic data (AN: N = 4425; BN: N = 4401; BED/PD: N = 4399; AAN: N = 4388; sBN/sBED = 4398; NES: N = 4220; UFED: N = 3989; Major ED: N = 4401; OSFED: N = 4223; Any ED: N = 4036). The proportions of each diagnosis do not add up to the totals, except for the other-Australian category of Major eating disorders because where the DSM-5 allows it there is comorbidity (for instance between the OSFED diagnoses) – this is why the major eating disorder category proportion total is closer to the summation of the individual diagnoses within that category (the DSM-5 does not allow concurrent diagnosis between AN, BN and BED). The chi-square analyses are weighted, which slightly alters cell numbers

Table 2 Univariate associations between demographic and clinical correlates and First Australian (FA) and eating disorder (ED) status, and their interaction

First Australian Other Australian	First Australian		Other Australian	ue				
	ED	No ED	Ð	No ED		Main Effect FA Status	Main Effect ED Status	Interaction Effect
	M (SD)				đ	F,η _p ² , p		
Age	15.0 (1.3)	14.4 (1.5)	15.2 (1.4)	14.7 (1.4)	1, 4045	7.35, .007, .002	25.40, .006, < .001	.19, .66, < .001
Socioeconomic status	975.0 (42.4)	965.1 (42.1)	987.4 (40.9)	988.3 (42.3)	1, 3984	40.11, < .001, .010	2.53, .001, .112	3.64, .057, .001
BMI percentile	69.2 (27.6)	55.7 (31.7)	61.6 (29.7)	50.7 (30.7)	1, 4045	9.95, .002, .002	36.69, .009, < .001	.41, .521, < .001
Weight and Shape Concerns	4.0 (1.8)	0.9 (1.2)	4.0 (1.6)	1.0 (1.2)	1, 4045	.60, .440, < .001	1235.15, .234, < .001	.02, .893, < .001
Psychosocial quality of life impairment	55.7 (27.2)	79.8 (20.4)	58.5 (25.2)	82.3 (19.0)	1, 3972	3.65, .056, .001	295.51, .069, < .001	.02, .896, < .001
Physical quality of life impairment	72.6 (25.6)	85.4 (22.9)	77.6 (23.4)	89.0 (15.5)	1, 3972	12.63, < .001, .003	101.16, .025, < .001	.35, .555, < .001
	(%) <i>u</i>					OR (95% CI) FA adolescent	OR (95% CI) Current ED	β (s.e.), <i>p</i>
Female Sex	59 (67.8)	84 (38.5)	593 (73.3)	1329 (45.5)		1.32 (0.81, 2.81)	3.30 (2.77, 3.93)	0.01 (0.29), .965
Objective binge eating	27 (31.0)	11 (5.0)	273 (33.7)	147 (5.0)		1.00 (0.54, 1.88)	9.54 (7.67, 11.87)	-0.12 (0.40), .766
Fasting	24 (27.6)	13 (6.0)	249 (30.9)	129 (4.4)		1.38 (0.77, 2.49)	9.68 (7.70, 12.17)	-0.47 (0.39), .225
Self-induced vomiting	15 (17.4)	5 (2.3)	90 (11.1)	25 (0.9)		2.88 (1.09, 7.61)	15.00 (9.51, 23.67)	-0.49 (0.58), .394
Laxative use	(0.7)	4 (1.8)	45 (5.6)	22 (0.8)		2.62 (0.89, 7.70)	7.80 (4.62, 13.17)	-0.72 (0.71), .311
Driven exercise	8 (9.2)	7 (3.2)	123 (15.2)	95 (3.2)		1.01 (0.46, 2.21)	5.40 (4.08, 7.13)	-0.61 (0.55), .266

Note. Socioeconomic status measured by SEIFA score, weight and shape concerns measured with the Eating Disorder Examination Questionnaire combined Weight and Shape Concern subscales score; quality of life Scale; objective binge eating, fasting, self-induced vomiting and laxative use defined as > 4 episodes in the past 28 days; driven exercise defined as > 20 episodes in the past 28 days. BMI body mass index

Burt et al. Journal of Eating Disorders (2020) 8:10 Page 7 of 10

Table 3 Multivariate associations between First Australian status, demographic and clinical correlates and eating disorder status

	Step 1	Step 2	Step 3	Step 4
	None	Socio-demographic	Socio-demographic + eating disorder symptoms	Socio-demographic + eating disorder symptoms + impairment
Variable	(Adjusted) Odds Rat	io (95% CI)		
First Australian status	1.44 (1.11, 1.87)	1.59 (1.20, 2.11)	1.55 (1.02, 2.35)	1.49 (0.97, 2.29)
Age		1.22 (1.16, 1.29)	1.08 (1.00, 1.17)	1.01 (0.93, 1.10)
Female sex		3.60 (3.03, 4.27)	0.67 (0.51, 0.86)	0.75 (0.57, 0.98)
SES		1.02 (0.85, 1.23)	0.95 (0.73, 1.25)	1.05 (0.80, 1.38)
BMI-percentile		1.14 (1.11, 1.17)	0.98 (0.94, 1.02)	0.99 (0.95, 1.03)
Weight/shape concerns			6.98 (6.04, 8.05)	6.18 (5.31, 7.20)
Objective binge eating			3.03 (2.25, 4.08)	2.95 (2.17, 3.99)
Self-induced vomiting			3.85 (2.01, 7.39)	2.48 (1.27, 4.86)
Laxative use			2.37 (1.03, 5.45)	2.89 (1.24, 6.78)
Fasting			2.03 (1.48, 2.79)	1.96 (1.41, 2.72)
Driven exercise			1.37 (0.91, 2.07)	1.55 (1.02, 2.37)
Psychosocial quality of life impairment				0.75 (0.65, 0.86)
Physical quality of life impairment				0.93 (0.82, 1.06)

Note. Significant AORs are in bold text

Reference categories for categorical variables: other-Australian status, male sex, < 4 episodes of objective binge eating/self-induced vomiting/laxative use/fasting, < 20 episodes of driven exercise

Scale for continuous variables: years (age), standard deviation unity (SES, weight/shape concerns, psychosocial quality of life impairment, physical quality of life impairment, deciles (BMI percentile)

with odds for current eating disorder, and this appeared to be largely explained by psychosocial quality of life impairment. Other factors that remained independently associated with meeting criteria for an eating disorder in the full multivariate model included all eating disorder symptoms and male sex.

Discussion

This study is the first to our knowledge to estimate the prevalence of eating disorders amongst First-Australian adolescents and supports previous research amongst First-Australian adults. Eating disorders are common amongst First-Australian adolescents. While overall, eating disorders were more common among First-Australian adolescents, the only specific diagnosis where First-Australian adolescents were statistically more likely to meet the diagnostic criteria was OSFED-NES, with other disorders being equally prevalent among First-Australian and other-Australian adolescents alike. The greater prevalence of eating disorders amongst First-Australian adolescents was largely explained by poorer psychosocial quality of life experienced more generally by First-Australian adolescents. This finding is in-line with eating disorder research suggesting poor quality of life is associated with eating disorders, and more specifically, recent research conducted with First-Australian adults indicating that mental health related quality of life impairment partly explained the higher prevalence of eating disorders among First-Australian adults. Eating disorders are well known to be associated with quality of life impairment [26]. There is evidence also that the link is bi-directional, in that while eating disorders impair quality of life, quality of life impairment in itself also increases the risk for developing eating disorders [27]. Thus, the poorer general quality of life among First-Australian adolescents may explain the greater prevalence of current eating disorders. First-Australians in this study also had lower socioeconomic status, which was also an explanatory variable for eating disorders.

As previously stated, there is a relatively larger pool of younger people in the First-Australian population compared to the other-Australian population, who are more socioeconomically disadvantaged and psychologically distressed, with poorer mental and physical health generally. In view of the similarity in findings of recent research conducted with First-Australian adults and adolescents, we suggest the higher prevalence of eating disorders amongst First-Australian adolescents and adults may be result of a relatively larger pool of people more vulnerable to mental disorders, including eating disorders. Given the differences between First-Australian groups culturally and the lack of research evidence, it is unclear whether specific cultural differences form part of the differences in eating disorder prevalence between First-Australians and other-Australians. It is unclear why OSFED-NES is specifically more common amongst

First-Australian adolescents, whereas other eating disorders are not. One potential mechanism may be a greater prevalence of sleep problems among First-Australian children [28]. OSFED-NES is linked to insomnia and the belief that one needs to eat to initiate or maintain sleep [29]. Clearly further research is required to confirm and to understand these potential, and other, explanations for the differences found.

Eating disorders amongst First-Australian adolescents were associated with weight and shape concerns. However, unlike research amongst First-Australian adults, these concerns were not markedly greater than in the other-Australians group in univariate analysis. A possible explanation for this is that unlike previous research, this study focused on adolescents, which is the peak period for body image concerns, with all youth at heightened risk. Combining the current findings with previous findings in adults, it could be the case that the decline in weight and shape concerns typically found following the adolescent risky period either does not occur, or occurs to a lesser extent among First-Australian youth (i.e., they experience a prolonged period of risk compared to their other-Australians counterparts).

In the multivariate analyses (Table 3) the findings of the step 3 model were the reverse of what was found in the step 2 model, which included just sociodemographic variables, and where female sex was associated with greater odds for current eating disorder. However, when eating disorder symptoms and impairment were added to the model, it appears that female sex may be associated with eating disorder status due to the higher rates of eating disorder symptoms and impairment among girls. These results suggest that when boys experience disordered eating, weight/shape concerns or impairment their likelihood of meeting criteria for an eating disorder is actually greater than that for girls. These data suggest that overvaluation may be a more benign phenomenon in girls, however, when experienced by boys, in particular when it is experienced by boys with mental health vulnerabilities, they are at increased risk of eating disorders.

Unless a patient presents underweight, eating disorders are easily overlooked as a cause of illness in the primary care setting, and given the high prevalence of eating disorders amongst First-Australians, there is a need to screen for eating disorders amongst First-Australians. An opportunity for this may be during Medicare Health Assessments for Aboriginal and Torres Strait Islander People (Medicare Benefits Schedule item 715), which are often conducted in Aboriginal Community Controlled Health Services (ACCHS) by General Practitioners. These are structured health assessments funded by the Australian Government's universal health scheme, known as Medicare. The cultural and language differences between First-

Australians and other-Australians, even amongst those in English-speaking communities suggests the need for a free-to-use, validated and culturally-specific instrument to assist clinicians in screening for eating disorders during these assessments. To our knowledge no such tool has been developed or validated for First-Australians. We suggest such a tool should be developed with First-Australian researchers and communities, similar to the development of the adapted Patient Health Questionnaire (aPHQ-9), a validated and culturally-specific screening tool for depression in First-Australians [29]. Finally, there is currently no evidence describing the uptake of eating disorder treatment for First-Australians. The introduction of Medicare Benefits Schedule items for eating disorders has also highlighted the need for education on eating disorders amongst stakeholders, such as General Practitioners, Nurses, Allied Health personnel and Aboriginal Health Workers, who are most likely to encounter undiagnosed eating disorders. Moreover, we suggest there should be increased awareness of eating disorders amongst school teachers, as they are also likely to encounter students with eating disorders.

This study had several strengths. It was based on data from the EveryBODY study, which was conducted over a broad geographical area including areas with sizable First-Australian populations and differing levels of socioeconomic advantage. The main limitation of this study was that the original data was not gathered with a deliberate intent to investigate eating disorders among First-Australians and not in consultation or collaboration with First-Australian community members or researchers. As such a culturally-informed approach to considering how eating disorders may present in First-Australians was not adopted at the design or data collection phases. On the other hand this specific study utilising the generated data has involved leadership and collaboration from First-Australian researchers (AB, LD), which has allowed for more cultural sensitivity in the interpretation of the existing literature and current findings. As stated above, of major importance in future research in this area would be the development of studies involving First-Australian researchers, that are also grounded within a First-Australian research paradigm as well as methodologies, such as yarning and community-based participation.

Conclusions

Based on our findings, eating disorders are common amongst First-Australian adolescents, and are associated with poor psychosocial quality of life. These findings are consistent with previous research conducted with First-Australian adults. Given eating disorders occur in the context of poor psychosocial quality of life, which may increase the risk of mental disorders generally, we suggest there is a need to screen for eating disorders in

Burt et al. Journal of Eating Disorders (2020) 8:10 Page 9 of 10

First-Australians. This process may be aided by the development of a validated, culturally specific screening tool for eating disorders in First-Australians. Finally, given the introduction of new MBS items for eating disorders, we suggest there is a need to increase education around the diagnosis and management of eating disorders, especially amongst First-Australians.

Supplementary information

Supplementary information accompanies this paper at https://doi.org/10. 1186/s40337-020-0286-7.

Additional file 1: Supplementary **Table 1.** Operationalisation of DSM-5 Eating Disorder Diagnoses (adapted from Mitchison et al. 2019)

Abbreviations

x²: Chi Squared test; AAN: Atypical Anorexia Nervosa; ABS: Australian Bureau of Statistics; ACCHS: Aboriginal Community Controlled Health Service; AMS: Aboriginal Medical Service; AN-B: Anorexia Nervosa Broad; aPHQ-9: Adapted Patient Health Questionnaire; ARFID: Avoidant Restrictive Food Intake Disorder; BED: Binge Eating Disorder; BMI: Body Mass Index; BN: Bulimia Nervosa; Cl: Confidence Interval; df: Degrees of freedom; DSM-5: Diagnostic and Statistical Manual of Mental Disorders 5th edition; ED: Eating Disorder; EDE-Q: Eating Disorder Examination Questionnaire; FA: First Australian; HOS: Health Omnibus Survey; ICD-11: International Classification of Disease 11th revision; MBS: Medicare Benefits Schedule; MWU: Mann Whitney U test; n: Number of participants; NES: Night Eating Syndrome; OR: Odds Ratio; OSFED: Other Specified Feeding or Eating Disorder; PD: Purging Disorder; PedsQL: Pediatric Quality of Life Scale; SBED: Subthreshold Binge Eating Disorder; SD: Standard Deviation; SEIFA: Socio-Economic Indexes For Areas; UFED: Unspecified Feeding or Eating Disorder; YLD: Years Lived with Disability

Authors' contributions

AB is a Wiradjuri (First-Australian) member of the research team, conceived the study, developed the research questions and was main author of the paper. DM conceived of the study with AB, conducted statistical analyses, and drafted the manuscript with AB. LD is a Worimi (First-Australian) member of the research team, assisted in drafting of the manuscript and data interpretation. PH conceived the study with AB, assisted with drafting the manuscript and data interpretation. KB, NT and AL assisted with drafting the manuscript and data interpretation. DM, KB, AL and NT collected and cleaned the data. All authors read and approved the final manuscript.

Funding

This study was supported by a Macquarie University Research Fellowship (awarded to DM) and Society for Mental Health Research Project Grant (awarded to DM).

Availability of data and materials

Data are available from the corresponding author for the purpose of secondary data analyses.

Ethics approval and consent to participate

The study was approved by the Macquarie University Human Research Ethics Committee (No: 5201918889546) and the New South Wales Department of Education (No: SERAP 2016357).

Consent for publication

Not applicable.

Competing interests

Professor Hay receives/has received sessional fees and lecture fees from the Australian Medical Council, Therapeutic Guidelines publication, and New South Wales Institute of Psychiatry and royalties/honoraria from Hogrefe and Huber, McGraw Hill Education, and Blackwell Scientific Publications, Biomed Central and PlosMedicine and she has received research grants from the

NHMRC and ARC. She is Chair of the National Eating Disorders Collaboration Steering Committee in Australia (2012-) and Member of the ICD-11 Working Group for Eating Disorders (2012–2018) and was Chair Clinical Practice Guidelines Project Working Group (Eating Disorders) of RANZCP (2012–2015). In the past 5 years she has prepared a report under contract for Shire Pharmaceuticals) and received honoraria for education of psychiatrists from Shire Pharmaceuticals (Takeda group). All views in this paper are her own.

Author details

¹School of Medicine, Western Sydney University, Sydney, Campbelltown, Australia. ²Translational Health Research Institute (THRI), School of Medicine, Western Sydney University, Sydney, Campbelltown, Australia. ³Centre for Emotional Health, Department of Psychology, Macquarie University, North Ryde, New South Wales, Australia. ⁴Ngarruwan Ngadju First Peoples Health and Wellbeing Research Centre, Australian Health Services Research Institute and Illawarra Health and Medical Research Institute, School of Psychology, University of Wollongong, Wollongong, New South Wales, Australia. ⁵Centre for Emotional Health, Department of Psychology, Macquarie University, Sydney, Australia. ⁶Campbelltown Hospital, SWSLHD, Sydney, Campbelltown, Australia.

Received: 7 December 2019 Accepted: 19 February 2020 Published online: 12 March 2020

References

- Al-Yaman F. Australian Burden of Disease Study: impact and causes of illness and death in Aboriginal and Torres Strait Islander people, 2011. Public Health Res Pract. 2017;27(4):56–67.
- Azzopardi PS, Sawyer SM, Carlin JB, Degenhardt L, Brown N, Brown AD, et al. Health and wellbeing of indigenous adolescents in Australia: a systematic synthesis of population data. Lancet. 2018;391(10122):766–82.
- Sanders RA. Adolescent psychosocial, social, and cognitive development. Pediatr Rev. 2013;34(8):354–8 quiz 8-9.
- Volpe U, Tortorella A, Manchia M, Monteleone AM, Albert U, Monteleone P. Eating disorders: what age at onset? Psychiatry Res. 2016;238:225–7.
- Mitchison D, Mond J, Bussey K, Griffiths S, Trompeter N, Lonergan A, et al. DSM-5 full syndrome, other specified, and unspecified eating disorders in Australian adolescents: prevalence and clinical significance. Psychol Med. 2019:1–10.
- Azzopardi PS, Kennedy EC, Patton GC, Power R, Roseby RD, Sawyer SM, et al. The quality of health research for young indigenous Australians: systematic review. Med J Aust. 2013;199(1):57–63.
- Black EB, Ranmuthugala G, Kondalsamy-Chennakesavan S, Toombs MR, Nicholson GC, Kisely S. A systematic review: identifying the prevalence rates of psychiatric disorder in Australia's indigenous populations. Aust N Z J Psychiatry. 2015;49(5):412–29.
- Hay P, Girosi F, Mond J. Prevalence and sociodemographic correlates of DSM-5 eating disorders in the Australian population. J Eat Disord. 2015;3:19.
- Mulders-Jones B, Mitchison D, Girosi F, Hay P. Socioeconomic correlates of eating disorder symptoms in an Australian population-based sample. PLoS One. 2017;12(1):e0170603.
- Burt AMJ, Mannan H, Touyz S, Hay P. Prevalence and features of eating disorders amongst aboriginal and torres strait islander peoples. Adelaide: Australia and New Zealand Academy for Eating Disorders (ANZAED) conference 2019; 2019.
- McCabe MP, Ricciardelli L, Mellor D, Ball K. Media influences on body image and disordered eating among indigenous adolescent Australians. Adolescence. 2005;40(157):115–27.
- The L. Indigenous health in Australia—moves in the right direction. Lancet. 2013;382(9890):367.
- 13. Department of Health. National Aboriginal and Torres Strait Islander health plan 2013–2023. Canberra: Canberra Australian Government; 2013.
- Trompeter N, Bussey K, Hay P, Mond J, Murray SB, Lonergan A, et al. Fear of negative evaluation and weight/shape concerns among adolescents: The moderating effects of gender and weight status. J Youth Adolesc. 2018; 47(7):1398–408.
- Census of Population and Housing: Socio-Economic Indexes for Areas (SEIFA), Australia, 2016. Canberra, ACT, Australia: Australian Bureau of Statistics 2018 [Available from: https://www.abs.gov.au/ausstats/abs@.nsf/mf/2033.0.55.001.

- Fairburn C, Beglin S. Eating disorder examination questionnaire (EDE-Q 6.0). Cognitive behavior therapy and eating disorders; 2008. p. 309–13.
- Mond J, Hall A, Bentley C, Harrison C, Gratwick-Sarll K, Lewis V. Eatingdisordered behavior in adolescent boys: eating disorder examination questionnaire norms. Int J Eat Disord. 2014;47(4):335–41.
- Nagl M, Jacobi C, Paul M, Beesdo-Baum K, Hofler M, Lieb R, et al. Prevalence, incidence, and natural course of anorexia and bulimia nervosa among adolescents and young adults. Eur Child Adolesc Psychiatry. 2016;25(8):903–18.
- Rojo-Moreno L, Arribas P, Plumed J, Gimeno N, García-Blanco A, Vaz-Leal F, et al. Prevalence and comorbidity of eating disorders among a community sample of adolescents: 2-year follow-up. Psychiatry Res. 2015;227(1):52–7.
- Allen KL, Byrne SM, Oddy WH, Crosby RD. DSM-IV-TR and DSM-5 eating disorders in adolescents: prevalence, stability, and psychosocial correlates in a population-based sample of male and female adolescents. J Abnorm Psychol. 2013;122(3):720–32.
- Hammerle F, Huss M, Ernst V, Burger A. Thinking dimensional: prevalence of DSM-5 early adolescent full syndrome, partial and subthreshold eating disorders in a cross-sectional survey in German schools. BMJ Open. 2016; 6(5):e010843.
- Allison KC, Lundgren JD, O'Reardon JP, Martino NS, Sarwer DB, Wadden TA, et al. The night eating questionnaire (NEQ): psychometric properties of a measure of severity of the night eating syndrome. Eat Behav. 2008;9(1):62–72.
- Gallant AR, Lundgren J, Allison K, Stunkard AJ, Lambert M, O'Loughlin J, et al. Validity of the night eating questionnaire in children. Int J Eat Disord. 2012;45(7):861–5.
- Varni JW, Seid M, Kurtin PS. PedsQL 4.0: reliability and validity of the pediatric quality of life inventory version 4.0 generic core scales in healthy and patient populations. Med Care. 2001;39(8):800–12.
- Varni JW, Burwinkle TM, Seid M, Skarr D. The PedsQL 4.0 as a pediatric population health measure: feasibility, reliability, and validity. Ambul Pediatr. 2003;3(6):329–41.
- Mitchison D, Morin A, Mond J, Slewa-Younan S, Hay P. The bidirectional relationship between quality of life and eating disorder symptoms: a 9-year community-based study of Australian women. PLoS One. 2015;10(3): e0120591.
- Blunden S, Chervin RD. Sleep, performance and behaviour in Australian indigenous and non-indigenous children: an exploratory comparison. J Paediatr Child Health. 2010;46(1–2):10–6.
- Allison KC, Lundgren JD, O'Reardon JP, Geliebter A, Gluck ME, Vinai P, et al. Proposed diagnostic criteria for night eating syndrome. Int J Eat Disord. 2010;43(3):241–7.
- Getting it Right Collaborative G. Getting it right: validating a culturally specific screening tool for depression (aPHQ-9) in aboriginal and Torres Strait islander Australians. Med J Aust. 2019;211(1):24–30.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions



Page 10 of 10